Author’s response to reviews

Title: Psycho-education programme for temporomandibular disorders: a pilot study

Authors:

Waseem Jerjes (waseem_wk1@yahoo.co.uk)
Geir Madland (waseem_wk1@yahoo.co.uk)
Charlotte Feinmann (waseem_wk1@yahoo.co.uk)
Stanton Newman (waseem_wk1@yahoo.co.uk)

Version: 2 Date: 11 March 2007

Author’s response to reviews:

Dear Editor

We would like to thank yourself and the reviewer for the constructive suggestions and corrections

Please note that all the recommended corrections and suggestions took place:

We have revised all the terminology and punctuation of the text

"Temporomandibular disorders (TMD)" corrected to become "Temporomandibular disorders (TMDs)" throughout the manuscript

"CR-ROM" corrected to become "CD-ROM" throughout the manuscript

A reference has been added at the end of line 1 of the introduction section as advised

"TMJ" corrected to be "Temporomandibular joint (TMJ)"

The "number of patients" in our study and all the other studies reviewed in our manuscript has been changed into numerical style as advised

"(RDC/TMD, Dworkin & LeResche, 1992)" have been changed to (RDC/TMD) [19], the reference has been added to the references list

There was a concern regarding the use of 2 terms (group and condition), we have dropped the term (condition) and used only one term (group) throughout the manuscript to make the manuscript easier to understand by the reader

The grammatical corrections in line 1 of the "Technical Report" took place as advised

We have changed the references according to the journal style

We have added 2 sections: competing interests and authors contributions according to the journal’s instruction to authors

Please do not hesitate to contact us if you have further queries

Kinds regard and thank you again

The authors